Author's response to reviews

Title: Does a self-referral counselling program reach doctors in need of help? A comparison with the general Norwegian doctor workforce.

Authors:

Karin E. I. Ro (k.e.i.ro@medisin.uio.no)
Tore Gude (tore.gude@medisin.uio.no)
Olaf G. Asland (olaf.aasland@legeforeningen.no)

Version: 4 Date: 15 March 2007

Author's response to reviews: see over
Vs "Core pre-accept":

To the BioMed Central Editorial Production Team.

Thank you for suggestions concerning formatting changes.

The running head has been removed.

Table 1 is revised according to your suggestions.

The text in the manuscript has been checked, with some small changes made concerning language issues.
The first part of the last paragraph in the discussion section on page 13 has been slightly revised – now describing more precisely the strengths of the study.

With sincere regards
Karin Rø